Author's response to reviews

Title: Selenoprotein S (SEPS1) gene -105G>A promoter polymorphism influences the susceptibility to gastric cancer in the Japanese population

Authors:

Tomoyuki Shibata (shibat03@fujita-hu.ac.jp)
Tomiyasu Arisawa (tarisawa@fujita-hu.ac.jp)
Tomomitsu Tahara (ttahara@fujita-hu.ac.jp)
Masaaki Ohkubo (mohkubo@fujita-hu.ac.jp)
Daisuke Yoshioka (dyoshioka@fujita-hu.ac.jp)
Naoko Maruyama (nmruyama@fujita-hu.ac.jp)
Hiroshi Fujita (hfujita@fujita-hu.ac.jp)
Yoshio Kamiya (ykamiya@fujita-hu.ac.jp)
Masakatsu Nakamura (mnakamura@fujita-hu.ac.jp)
Mitsuo Nagasaka (mnagasaka@fujita-hu.ac.jp)
Masami Iwata (miwata@fujita-hu.ac.jp)
Kazuya Takahama (ktakahama@fujita-hu.ac.jp)
Makoto Watanabe (mwatanabe@fujita-hu.ac.jp)
Ichiro Hirata (ihirata@fujita-hu.ac.jp)

Version: 5 Date: 6 January 2009

Author's response to reviews: see over
Dear Dr Dunckley,

Thank you very much for reviewing our manuscript once again entitled “Selenoprotein S (SEPS1) gene -105G>A promoter polymorphism influences the susceptibility of Japanese to gastric cancer” (MS: 1983473344218508). We are very pleased to learn that our manuscript has been accepted for publication in BMC gastroenterology. We revised our paper according to editorial production team’s request. Please check again our manuscript for publication in BMC gastroenterology.

Sincerely yours,

Tomoyuki Shibata, M.D., Ph.D.